Author's response to reviews

Title: Extracorporeal shock waves down-regulate the expression of interleukin-10 and tumor necrosis factor-alfa in osteoarthritic chondrocytes

Authors:

Biagio Moretti (b.moretti@ortop2.uniba.it)
Florenzo Iannone (f.iannone@reumbari.uniba.it)
Angela Notarnicola (angelanotarnicola@yahoo.it)
Giovanni Lapadula (g.lapadula@reumbari.uniba.it)
Lorenzo Moretti (b.moretti@ortop2.uniba.it)
Vittorio Patella (v.patella@ortop2.uniba.it)
Raffaele Garofalo (raffaelegarofalo@hotmail.com)

Version: 6 Date: 11 December 2007

Author's response to reviews:

Dear Editor,

we provide the name of the Commettee who gave approval for our study: Polyclinical Ethical Commettee of Bari.

We look forward your following news about our manuscript,

thank you,

all the authors